Author’s response to reviews

Title: Potentiating and synergistic effect of grape fruit juice on the antioxidant and antiinflammatory activity of aripiprazole against hydrogen peroxide induced oxidative stress in mice

Authors:
Seema Zargar (szargar@ksu.edu.sa)
Tanveer Wani (twani@ksu.edu.sa)
Abdul-Rahman Al-Majed (tanylak@yahoo.co.in)

Version: 5 Date: 28 Sep 2017

Author’s response to reviews:

All the other reviewers have accepted the manuscript as such except Dr. Ratana Banjerdpungchai who suggested removal of figure 6 and 7.

Comment: The results in Figure 6 and 7 could not bring any significant and the authors might delete them and without make the conclusion changed.

Response: As per the esteemed reviewers suggestion both the figures have been removed from the revised manuscript.